- We thank all reviewers for their insightful comments. 1
- We left as an open problem in the paper for extending the convergence result to general multiplayer games when everyone runs Optimistic Hedge. Our current analysis does not work for three-player games due to some technical 2
- 3
- 4 difficulties. But we believe that some of the technical ingredients and insights developed in this work could be useful
- ⁵ for future research on this problem.